Author's response to reviews

Title: Perforated Sigmoid Diverticulitis in a Lumbar Hernia after Iliac Crest Bone Graft - a Case Report

Authors:

Florian S Frueh (Florian.Frueh@usz.ch)
Raphael N Vuille-dit-Bille (raphael.vuille@access.uzh.ch)
Dimitri A Raptis (draptis@btinternet.com)
Hanspeter Notter (hanspeter.notter@spitalbuelach.ch)
Brigitte S Muff (Brigitte.Muff@spitalbuelach.ch)

Version: 4 Date: 11 July 2014

Author's response to reviews: see over
**Cover letter giving a point-by-point response to the concerns.**

Title Page
Please remove all author suffixes. This information may be included in an authors’ information section if desired:
http://www.biomedcentral.com/bmcmusculoskeletdisord/authors/instructions/researcharticle#
formatting-information
Please remove your keywords and abstract from the title page. More information on formatting your title page and manuscript may be found here:
http://www.biomedcentral.com/bmcsurg/authors/instructions/casereport

*This was done. All suffixes were removed. Keywords and abstract were removed from the title page.*

Abstract
Please check that your abstract contains all the required information:
http://www.biomedcentral.com/bmcsurg/authors/instructions/casereport#abstract
Please ensure that you update your abstract in both the submission system and in the manuscript itself.

*All necessary informations (patient’s age, gender, ethnicity) are now given in the abstract.*

Case Presentation
Please include information about the patient’s ethnic background in your case presentation. This should also be included in your abstract.

*This was done as well.*